Author's response to reviews

Title: An osseous lesion at presentation of Hodgkin lymphoma in a 10-year old boy: a case report

Authors:
Machiel van den Akker (vdakker@bgu.ac.il)
Vadiem Zudekov (vadim567@zhav.net.il)
Asher Moser (amoser@bgu.ac.il)
Joshep Kapelushnik (JosephK2@clalit.org.il)

Version: 3 Date: 22 June 2011

Author's response to reviews:

June 22, 2011
Michael Kidd, Editor-in-Chief
Journal of Medical Case Reports

Dear Dr Michael Kidd,

Please find enclosed the original copy of our manuscript entitled ‘An osseous lesion at presentation of Hodgkin disease in a 10-year old boy: a case report’, by Machiel van den Akker et al., which was send for publication in the Journal as a brief report.

As requested by the reviewers some revisions were made:
- in the abstract emphasize in stead of emphasizing
- in the discussion paragraph 1 line 10 an MRI in stead of a MRI
- author’s contribution first line: data collection in stead of collection of data
- in the case presentation second paragraph line 5 periosteal in stead of periostal
- ethnicity was added in the abstract and the case presentation as well
(as requested all the changes are marked in red)

The involved patient is a minor, therefore written consent was received from the parents for publication of the case.

Sincerely Yours,
Machiel van den Akker

Previous: University Medical Center Soroka, Israel
Current: Queen Paola Children Hospital, Antwerpen, Belgium
Phone: +3232802134
e-mail address: vdakker@bgu.ac.il